Author's response to reviews

Title: Influence of low dose Aprotinin on thoracic surgical operations: A prospective randomized trial

Authors:

Efstratios Apostolakis (stratisapostolakis@yahoo.gr)
Nikolaos Panagopoulos (ekoletsis@upatras.gr)
Efstratios Koletsis (ekoletsis@hotmail.com)
Jim Crockett (fccp_perfusionist@yahoo.co.uk)
Helen Stamou-Kouki (elenistamou21@hotmail.com)
Efrosini Sourgiadaki (ekoletsis@upatras.gr)
Kriton Filos (ekoletsis@upatras.gr)
Dimitrios Dougenis (ddougenis@med.upatras.gr)

Version: 2 Date: 3 March 2008

Author's response to reviews: see over
Thank you for reviewing this paper for publication. The authors are happy to admit that all the comments from your reviewers were positively constructive. An extensive editing procedure took place and many amendments have been performed in the manuscript. In an effort to respond to the issues raised we hope that we transformed this paper in such a way that it will provide a better understanding of our results and our message more clear. In order to facilitate the reviewing process we will attempt to address the issues raised by each reviewer separately in a similar way to the one we received the comments. Please excuse the necessary repetition in certain cases.

Specifically, the following points from editor were addressed:

**rev 1**

*Did the authors mention any renal complications in their study?*

No we did not had any renal complications

*Do they also have any experience with the use of Tranexamic acid (TRANSAMINE)?*

Only in cardiac operations

*Please clarify whether other hemostatic regimes have been used during the study period.*

No other haemostatic regiment was used.

*Needs some language corrections before being published*

Your request has been respect

**rev 2**

*There are a few typographical errors, which they should be corrected prior to publication*

Your advice has been respected
Thank you in advance

Efstratios Koletsis MD, PhD
Senior Lecturer
University of Patras